Author's response to reviews

Title: HMGA1 and HMGA2 expression and comparative analyses of HMGA2, Lin28 and let-7 miRNAs in oral squamous cell carcinoma

Authors:

Katharina A. Sterenczak (Katharina.Sterenczak@tiho-hannover.de)
Andre Eckardt (Eckardt.Andre@mh-hannover.de)
Andreas Kampmann (Kampmann.Andreas@mh-hannover.de)
Saskia Willenbrock (Saskia.Willenbrock@tiho-hannover.de)
Nina Eberle (Nina.Eberle@tiho-hannover.de)
Florian Länge (Laenger.Florian@mh-hannover.de)
Sven Kleinschmidt (Sven.Kleinschmidt@tiho-hannover.de)
Marion Hewicker-Trautwein (Marion.Hewicker.Trautwein@tiho-hannover.de)
Hans Kreipe (Kreipe.Hans@mh-hannover.de)
Ingo Nolte (ingo.nolte@tiho-hannover.de)
Hugo Murua Escobar (hescobar@tiho-hannover.de)
Nils C. Gellrich (Gellrich.Nils-Claudius@mh-hannover.de)

Version: 2 Date: 2 June 2013

Author's response to reviews: see over
Dear Editor, dear Editorial Team,

please find enclosed the revised version of our BMC Cancer manuscript: MS# 1457729371953663.

In this revised manuscript version we added the information concerning the ethical approval of the human and canine samples.

Thank you very much in advance for your consideration.

With my best wishes from Hannover,

H Murua Escobar
List of Changes

BMC Cancer Manuscript: 1457729371953663; Sterenczak et al.

Statements added:

Page 8:

Paragraph, 2. Material and Methods, (2.1. Tissue samples obtained from human patients)

The following sentence was added:

“This study was approved by the local ethics committee at the Hannover Medical School. (Ref No. 984-2011).”

Paragraph, 2. Material and Methods, (2.2. Tissue samples obtained from canine patients)

The following sentence was added:

“All samples were taken and provided by the Small Animal Clinic, University of Veterinary Medicine, Hannover, Germany according to the legislation of the state of Lower Saxony, Germany.”